



Article ABANICCO: A New Color Space for Multi-Label Pixel Classification and Color Analysis

Laura Nicolás-Sáenz ^{1,2}, Agapito Ledezma ³, Javier Pascau ^{1,2}, and Arrate Muñoz-Barrutia ^{1,2,*}

- ¹ Departamento de Bioingeniería, Universidad Carlos III de Madrid, 28911 Leganes, Spain; lnicolas@ing.uc3m.es (L.N.-S.)
- ² Instituto de Investigación Sanitaria Gregorio Marañón, 28007 Madrid, Spain
- ³ Departmento de Informática, Universidad Carlos III de Madrid, 28911 Leganes, Spain
- Correspondence: mamunozb@ing.uc3m.es

Abstract: Classifying pixels according to color, and segmenting the respective areas, are necessary steps in any computer vision task that involves color images. The gap between human color perception, linguistic color terminology, and digital representation are the main challenges for developing methods that properly classify pixels based on color. To address these challenges, we propose a novel method combining geometric analysis, color theory, fuzzy color theory, and multi-label systems for the automatic classification of pixels into 12 conventional color categories, and the subsequent accurate description of each of the detected colors. This method presents a robust, unsupervised, and unbiased strategy for color naming, based on statistics and color theory. The proposed model, "ABANICCO" (AB ANgular Illustrative Classification of COlor), was evaluated through different experiments: its color detection, classification, and naming performance were assessed against the standardized ISCC-NBS color system; its usefulness for image segmentation was tested against state-of-the-art methods. This empirical evaluation provided evidence of ABANICCO's accuracy in color analysis, showing how our proposed model offers a standardized, reliable, and understandable alternative for color naming that is recognizable by both humans and machines. Hence, ABANICCO can serve as a foundation for successfully addressing a myriad of challenges in various areas of computer vision, such as region characterization, histopathology analysis, fire detection, product quality prediction, object description, and hyperspectral imaging.

Keywords: image color analysis; image analysis; semantics; fuzzy color space; color modeling; color segmentation; color classification; human perception

1. Introduction

Color plays a crucial role in computer vision: thus, color analysis is essential to tasks across various industries, such as medicine, satellite imaging [1], food quality assessment [2], visual scene description, and city planning. Despite its importance, color analysis remains a challenging problem, due to the complex nature of color, the lack of clear transitions or limits, the bias of human perception, the effects of context, and the semantic gap between subjective human perception and computer color representation.

Throughout history, there have been many attempts to deal with the complexities of color representation and analysis. Color has been represented in various ways, including the traditional color wheel and, in the age of computer vision, using color spaces, such as RGB and CIELAB. Computer vision is the branch of computer science that deals with the many challenges of color analysis and representation. A color vision algorithm should be able to reproduce the human experience of color perception, and to identify the colors present in an image, giving them a color name or color term. Additionally, the algorithm should be able to locate these color terms within a digital color space, and to use these subspaces for accurate color classification and image segmentation. However, several



Citation: Nicolás-Sáenz, L.; Ledezma, A.; Pascau, J.; Muñoz-Barrutia, A. ABANICCO: A New Color Space for Multi-Label Pixel Classification and Color Analysis. *Sensors* **2023**, *23*, 3338. https://doi.org/10.3390/s23063338

Academic Editors: Jean-Baptiste Thomas, Guangyuan Wu, Qiang Liu and Xufen Xie

Received: 10 February 2023 Revised: 15 March 2023 Accepted: 19 March 2023 Published: 22 March 2023



Copyright: © 2023 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). problems exist that render the creation of such a perfect color analysis algorithm difficult. The main problem with color vision is that human perception of color is not homogeneous: each person has their own experience of color, influenced by context, culture, education, and lighting [3,4]. Furthermore, human perception is also influenced by the phenomenon of color constancy, by which our brains perceive color homogeneously on surfaces, despite possible changes of lights or reflections of brighter colors, in contrast to the pixel-level differences in color detected by computers [5]. Thus, it is impossible to create a computer method able to reproduce the infinite possible experiences of color. The second problem, influenced by the differences in human perception of color, is that color names are not universal. The same shade of red could be called "crimson" or "cherry", depending on the observer [6]; therefore, there is a great disparity between linguistic descriptions of color and computer representations. This problem is known in the literature as 'the semantic gap' [7]. The ultimate challenge for color vision algorithms is that there are no well-defined boundaries between color categories, i.e., there are no immediate methods of finding, for example, the limit between orange and red, or between yellow and green. Humans will, in fact, resort to compound names (reddish-orange, yellowish-green) when confronted with colors located in fuzzy areas between more established terms [8].

Many alternatives have been introduced to deal with these problems, in order to create accurate computer vision methods for color detection and classification. The current research on color analysis primarily focuses on two areas: (1) color naming and detection; (2) color segmentation. Studies on color naming and detection aim to find a mapping between digital color representation and human language for color: this is a complex task, because of the subjective nature of human understanding of color [9]. Alternatively, fuzzy models, which handle the ill-defined boundaries between colors, are able to close the semantic gap while also performing well in the second area: color segmentation [8]. Another approach to color segmentation is color quantization, which is based on clustering, but it is limited by dependence on the training set [10,11].

In this paper, we introduce ABANICCO (AB ANgular Illustrative Classification of COlor). ABANICCO is a novel approach to color detection, naming, and segmentation that addresses the above limitations, and provides more accurate analysis of color. The proposed method uses color theory, fuzzy logic principles, and multi-label systems to create an efficient model that can perform the aforementioned tasks automatically, making it suitable for real-world applications. The main highlights of our method are:

- ABANICCO is an easy-to-use, automatic, non-supervised method for color analysis segmentation, classification, and naming;
- it is based on objective color theory, thus eliminating human color bias, and making the analysis trustworthy and reproducible;
- it incorporates fuzzy logic concepts to approximate color naming to human linguistics and understanding, thus bridging the semantic gap;
- finally, ABANICCO allows the user to make informed decisions, and to modify the obtained results, to fit specific applications: this is possible thanks to ABANICCO's ease of use and excellent graphical representation of the detected colors.

The remainder of the paper is organized as follows: Section 2 introduces some fundamental concepts of color analysis, which are necessary in order to comprehend the problem—with a particular emphasis on those aspects that are most pertinent to the research proposed here; Section 3 presents the state of the art, and its limitations; in Section 4, we introduce ABANICCO and its innovations, with respect to the state-of-the-art methods; Section 5 relates the steps taken to build our model; Section 6 shows the experimentation with the proposed approaches, and their results; finally, Section 7 summarizes our work. Additionally, we include a Graphical Abstract and in Appendix A, extra technical information about the steps in the creation of ABANICCO.

2. Fundamentals

2.1. Color Theory

Color theory is the science that explains how humans perceive color: this perception results from how light is absorbed, reflected, and transmitted by the objects around us. When light strikes an object, some wavelengths are absorbed while others are reflected: this reflection is what allows us to see the color of an object. Nevertheless, all humans have different cultural and contextual associations with color that vary our perceptions [12–14]. As Newton said, "Colour is a sensation in the viewers' mind" [15].

Three color-making attributes are used to communicate this subjective experience of color: hue; hue purity (expressed as colorfulness, chroma, saturation, or dullness); and lightness or darkness. In turn, hues are divided into primary, secondary, and tertiary. The primary hues are Red, Yellow, and Cyan. Blending two primary hues gives the secondary ones: Green, Orange, and Purple. Lastly, we obtain tertiary hues by mixing a primary hue with its nearest secondary. The six tertiary hues are Red-Orange, Yellow-Orange, Yellow-Green, Blue-Green, Blue-Violet, and Red-Violet. However, the nature of color is continuous; therefore, the best representation of colors is the color or hue wheel created by Sir Isaac Newton in 1704 [16]. Figure 1A shows how this hue wheel can be understood as the progressive blending of the primary, secondary, and tertiary colors represented inside it. It is important to note that some colors are not represented in the color wheel, because the color wheel only shows pure or fully chromatic colors. Modifications to these pure hues create the rest of the possible colors. The hues can be modified by changing the color-making attributes of purity and lightness with achromatic or neutral colors—white, black, and gray—to obtain tints, shades, and tones, respectively. A wider variety of colors can be seen in the tints, shades, and tones images of Figure 1B. This difference between pure hues and modifications can be better understood if we single out certain shades and tones (Figure 1D). Triangle 5.t shows what can be referred to as "forest green"; however, this deep shade of green is usually described simply as "dark green". Circles 1 to 4 show increasingly dark (1–4.s) shades and duller tones (1–4.t) of orange: when observed separately, it is clear that these colors are what we usually refer to as "brown." Brown is technically a near-neutral dark-valued orange; however, in contrast to dark green, almost nobody would refer to brown as dark orange: this is a crucial point to understand, because most research conducted to date on color detection and classification aims to find a unique space for brown as a pure hue instead of treating it as a modified hue, the same as dark green or dark blue [17–19]. This repeated inaccuracy in color research—which contradicts even Goethe's and Newton's precise placement of brown as dark orange in their color wheels [20,21]—likely stems from the complicated gap between color theory, color perception, and color naming [22].

2.2. Color Terms

The continuous nature of color, with no clear transitions or limits, makes classifying and naming individual colors problematic. The first attempts at color naming and classification were driven by the boom of exploratory trips and expeditions in the 18th and 19th centuries ([18,19,23–25]); however, all these proposals relied heavily on context and cultural cues. In the 20th century, two significant attempts were made at color naming standardization, and remain the baseline against which most research is compared:

- i. To standardize color naming, the Inter-Society Color Council (ISCC) and the National Bureau of Standards (NBS) first established, in the 1930s, the ISCC–NBS System of Color Designation [26], which defines three levels of color naming: Level 1 consists of 10 color terms; Level 2 adds 16 intermediate categories; and Level 3 uses brightness and pureness modifiers to obtain 267 color categories. This system was used in one of our experiments, to test the accuracy of ABANICCO's color naming in Section 6.1.
- ii. In 1969, Berlin and Kay examined the effect of culture on the differences in color naming in natural languages. They found 11 basic color categories in most of them: white, gray, black, yellow, orange, red, pink, purple, blue, green, and brown. In their

work *Basic colour Terms: Their Universality and Evolution* [27], they were able to link the universality of these terms to similar evolutionary processes. They also found that within these categories, the central regions would elicit higher consensus than the boundaries, where consensus tapered off, consistent with the continuous nature of color.



Figure 1. Color theory bases: (**A**) representation of color wheel; the color wheel is obtained from the primary colors (P) Red, Yellow, and Blue; by blending the primary colors, we get the secondary (S) colors, Green, Purple, and Orange; finally, by blending the secondary colors, we obtain the tertiary (T) colors, Yellow-Green, Blue-Green, Blue-Violet, and Red-Violet; (**B**) depiction of the concept of tints, shades, and tones with the color wheel in A; in every wheel, each ring going outward shows a 25% increase in added white for tints, black for shades, and gray for tones; these show the near-achromatic colors. (**C**) the four main color spaces used in digital imaging—RGB, YCbCr, HSV, and CIELAB; (**D**) color picker results displaying the colors within the shapes marked in the shades (s)- and tones (t)-modified color wheels; the circles show the effect of adding black (shades) or gray (tones) to orange; the triangle shows a dark shade of green called forest green; the obtained results show that what we understand as brown is not a pure color but rather shades of pure colors ranging from red to warm yellow.

Berlin and Kay's categorization and the ISCC–NBS System are the most-used color terms standardizations in digital imaging and color research nowadays. However, both present approaches have problems that limit their practical use. Berlin and Kay's color terms fail to combine successfully to create secondary and intersectional classes. Additionally, the experiments were conducted using exclusively high chroma levels, never testing less saturated colors. Furthermore, the ISCC–NBS color terms may seem arbitrary to different users who do not agree with the provided descriptions. Moreover, the terms on Levels 2 and 3 do not follow logical reasoning that could make them understandable to any human or machine, as demonstrated by Anthony E. Moss in [28].

2.3. Color Models

Due to the difficulties in finding appropriate, accordant names for colors, color models were introduced throughout the 20th century. Color models are systems that describe colors using numerical values instead of linguistic terms. The most influential one is the Munsell color system, created in 1913 by Professor Albert H. Munsell as a rational way to describe colors [29]. The fundamental premise of the Munsell color system is the distribution of colors along three perceptually uniform dimensions: hue; value; and chroma. The placement of the colors was decided by measuring human responses, instead of trying to fit a predefined shape, thus obtaining an irregularly shaped cylindrical space that avoids distorting color relations [30]: in this space, hue is measured in degrees around horizontal circles, chroma goes radially outwards from the neutral vertical axis, and the value increases vertically from 0 (black) to 10 (white). The irregular shape of the Munsell color space allows for wider ranges of chroma for specific colors, which follows the physics of color stimuli and the nature of the human eye. This non-digital color system is the most-used in psychology and soil research [31,32].

With the rise of computer operations, digital color models were created to suit the new tasks better. The main digital color spaces are RGB, HSL/HSV, and CIELAB. A visual representation of these can be seen in Figure 1C. RGB was created in 1996 for internet and display purposes: in this space, colors are represented as three coordinates corresponding to red, green, and blue values. Despite being the most widely used color space in digital applications, the colors it can represent are limited, device-dependent, and discordant from human perception and interpretation. By contrast, HSL/HSV spaces were designed circularly, to better align with human perception of color-making attributes: these spaces consist of a three-dimensional cylindrical transformation of the RGB primaries into a polar coordinate system, obtaining hue, saturation, and value (in HSV) or lightness (in HSL). Finally, CIELAB is a uniform color space, based on human perception, that achieves the best compromise between complexity and accurate color representation by adopting the concepts of the Munsell color system. In CIELAB, the geometrical distance between any two points represents the perceived color differences without forcing the color relationships into strict geometrical shapes: this makes CIELAB more perceptually linear and decorrelated than the alternatives [33–35].

3. Related Work

Color analysis poses numerous challenges, including variations in perception, limited linguistic mapping between language and machines, and lack of clear boundaries, as explained in the previous section. Currently, color analysis research primarily focuses on resolving two related problems: (1) color naming and detection; and (2) color segmentation.

Much research has been devoted to color naming and detection, with the aim of finding a suitable mapping between the digital representation of colors and the language employed by humans when referring to them to bridge the semantic gap [6,9,36]. However, this is a complex task due to the subjective nature of human understanding of color, which can vary, based on factors like age, context, profession, and eye morphology. The only way to truthfully close the semantic gap would be to run a multitudinary experiment with a sufficiently heterogeneous cohort that would consider every possible experience of color: however, this would be highly impractical.

Fuzzy models, which tackle the notoriously ill-defined boundaries between colors, are being used as a practical alternative. Fuzzy set theory is closer to the human cognitive process of color categorization than crisp color partitions [7]: because of this, several research works have applied fuzzy logic to successfully obtaining a correspondence between human perception, color language, and computational color representation. Within this approach, three subgroups of methods stand out: (1) those that rely on geometrically regular membership functions; [2,37,38]; (2) those that formalize fuzzy color spaces from human-obtained prototypes [6,36]; and (3) those that create adapted fuzzy color spaces dependent on application [8,39–42]. Methods in the first subgroup are problematic, because

they rely on color spaces with arbitrary distances between color categories. Moreover, using regular, homogeneous functions disregards the irregular form of color terms [43]. Methods in the second subgroup provide good color models, but depend on the number of study participants, and focus on a limited number of color terms obtained from subsets of a single cultural environment (like Berlin and Kay). Methods in the third subgroup favorably address the limitations of the previous methods, and achieve good results in color segmentation, as well as in color detection and naming: however, the resulting pipelines are complex and inaccessible for regular use, and are challenging to adapt to new applications.

In addition to fuzzy color models, color quantization is widely used for color segmentation. Color space quantization is based on clustering, usually by nearest neighbor classification or color thresholding. Classification based on nearest neighbors finds the *k* closest neighbors to any given pixel in a set of predefined color classes [10,11]: this approach is limited by the necessary definition of the clusters and boundaries, which depends entirely on the training set, although some recent research provides more robust results [44]. Alternatively, color thresholding divides the space into segments defined by mostly-linear boundaries that need to be manually decided: this step has been solved using machine learning techniques, which require a priori knowledge of the labels of the color categories and training process, or by complex deep learning optimization algorithms [45–47]. Like the previous method, this approach is severely limited by the dependence on training data [10,11].

Nevertheless, the problem most previous methods have in common is disregarding color theory fundamentals. Excepting the most recent fuzzy model approaches, most of the above-mentioned approaches are based on misinformed notions of what colors are: these inaccuracies are often due to the confusion between pure hues and color terms. Basic Color Terms (BCTs, as introduced by Sturges & Whitfield in 1991) include 11 categories: Black; White; Red; Green; Yellow; Blue; Brown; Orange; Pink; Purple; and Gray [27,43]. Since then, many research works have aimed to locate these color terms in the color spaces, as if they were pure hues: this becomes particularly problematic in the case of brown [36], for the reasons stated in the previous section.

4. Proposal

This paper introduces ABANICCO, a new approach to pixel classification, color segmentation, and naming. Our proposal boasts groundbreaking innovations in the field. Firstly, we have formalized the notion of our modified AB polar, two-dimensional color space as the primary object for color analysis. This innovative representation offers a clear graphical understanding of the colors present in an image, and of how they are organized. Our modified color space is divided into 12 categories: Pink; Red; Red-Orange; Yellow-Orange; Yellow; Green; Teal; Blue; Ultramarine; Purple; Brown; and Neutral. These categories were determined by color theory, thus avoiding the need for tedious experiments and a large number of experiment subjects.

Secondly, we present a methodology that utilizes fuzzy principles and multi-label systems to name the identified colors, bridging the semantic gap by building on fuzzy color models. The functions used are based on the color categories established through color theory, and consider irregular color shapes in line with concepts outlined in prior research [29,43].

As a third innovation, our color mapping allows for accurate color segmentation. As we do not rely on human-made labels or annotations, our color segmentation is objective; moreover, it avoids problems such as the effect of light, issues with perception, the phenomenon of color continuity, and the effects of context and personal bias.

These innovations are visually summarized in the graphical abstract included.

Finally, we propose a simple pipeline for image segmentation, based on the color mapping provided by ABANICCO. The main advantage of this pipeline over existing approaches is the possibility of the user making a fully informed decision about the areas to segment, thanks to the polar description ABANICCO provides of the images.

5. Methods

To create the proposed method, we approached the problem of color analysis in two steps. Firstly, we solved the problem of automatic color identification and segmentation, by using color theory and simple geometry concepts to study the color wheel, and to obtain the pure hue bases in the CIELAB color space. We named this step "Geometric knowledge generation." Secondly, we used fuzzy logic notions, color theory and multilabel classification to name and describe the identified colors: this step was called "Fuzzy space definition and multi-label classification for colour naming and description".

5.1. Geometric Knowledge Generation

The purpose of this step was to create a simplified color space for color classification and naming. We chose to work with CIELAB color space, due to its advantageous properties, as explained in Section 2.3.

The first task was to find the exact location of the pure hues in the color wheel within the CIELAB color space. As color space dimensions are decorrelated, the luminance component is unnecessary for understanding a given color's chromatic content: this meant that we could discard the luminance dimension, and work on the reduced AB two-dimensional space. To find the location of the main crisp or pure hues in this reduced color space, we selected an example of a complete color wheel, including hue modifications. We calculated the double AB histogram, showing the location of the crisp hues. The resulting geometrical distribution radiated from a center point, as shown in Figure 2A. The skeletonization of the double histogram provided endpoints and branches following a circular path. Due to their disposition, the lines between the center of the dual histogram and each endpoint could be used to represent the bases of the pure chromogens. Figure A1 provides a better look at this representation.

Once we knew where the pure bases were, the next task was to find the boundaries between them. We considered two options for locating the boundaries between the crisp color categories: angle bisectors and classification methods. Both were tried out before moving to the next step, to decide on the best approach. The classification methods were more complex to use, needed training, and took longer to provide results, and the obtained classifications were unsatisfactory, both in resulting boundaries and classes created. An example of these unsatisfactory boundaries applied to natural images was the obtained segmentations of a gradient of colors in Figures A2 and A3. We observed how the machine learning classifiers failed to create appropriate boundaries between the three present colors, each generating a single class of arbitrary length instead of the expected classes of set length obtained by the bisectors approach: these results, together with the simplicity and efficiency of the bisectors approach, explain why we chose it over other classification methods. The experiments carried out to determine the best method are described in Appendix A, and the obtained segmentation of the whole range of hues in Figures A2 and A3.

Using angle bisectors, we obtained precise color regions that, for simplicity, were then grouped according to color perception, shape, CIELAB construction, and the Berlin and Kay theories [27,33], in 10 groups: Green; Yellow; Light Orange; Deep Orange; Red; Pink; Purple; Ultramarine; Blue; and Teal.

It is important to note that neutrals (black, white, and gray) and near-neutral brown were not present in this classification, as they are not pure chromatic colors. In the CIELAB color space, the pure hues are located far from the center, and are radially distributed. All modifications of these hues then fall gradually closer to the center, with the shades having lower luminance than the tints, and the achromatic colors falling in a gradient aligned on the L axis. This can be seen in the CIELAB diagram of Figure 1C. In our reduced AB space, this meant that the neutral colors, black, white, and gray were located in the center of the space. The shades and tones of orange, commonly named brown, were located close to the center, within the red, orange, and yellow areas. Figure 2 shows how this area corresponded to the lower left corner of the upper right quadrant in the AB space (consistent with the mixed range of yellows and reds), close to the L axis (due to their

near-achromacity); therefore, brown shades could be aligned with any pure chromogen bases between red and warm yellow, depending on the angle and distance from the center.



(A) Geometric Knowledge Generation





Figure 2. The two main steps of ABANICCO: (**A**) geometric knowledge generation; we used color theory to identify in the color wheel the localization of the different pure hues, shades, and tints within the reduced AB space of CIELAB; applying geometry, we used bisectors as the best boundaries between pure hues; by these means, we obtained a discrete polar color space divided into 12 different color categories—Pink, Red, Red-Orange, Yellow-Orange, Yellow, Green, Teal, Blue, Ultramarine, Purple, Brown, and Achromatic; the first 10 categories described hue, depending on the angle; the last two described chromaticity, and depended on the radius; (**B**) fuzzy space definition and multi-label classification for color description; we formalized fuzzy rules to translate the obtained discrete color polar space back into a continuous space that better mirrored human linguistics; based on the bisectors employed for the discretization of the space, we were able to define areas of absolute chromogen certainty and gradients of membership to the adjacent colors on both sides, which was done for both the radius (accounting for chromaticity) and the angle (accounting for hue); finally, with color theory, fuzzy membership functions, and multi-label classification, we assigned multiple, non-exclusive labels to each detected color, for accurate naming and description.

The final task was the incorporation of browns and neutral as two extra classes. For this purpose, we included two additional areas in our simplified color space, which depended on the distance from the center: the first area was the circular area of radius r_1 (Critical Achromatic Radius) corresponding to the neutral pixels centered on the L axis—the whites, grays, and blacks in the image; the second area was a quadrant of radius r_2 (Critical Near-Achromatic Radius) corresponding to the positive values of A and B in the 2D color space, and comprising all shades of brown. The radii r_1 and r_2 corresponded to the first and second sets of branching points in the skeletonized double histogram, as seen in Figure A1 of Appendix A.

In this manner, a new, simplified color space was created, capable of classifying colors into 12 categories: Pink; Red; Red-Orange; Yellow-Orange; Yellow; Green; Teal; Blue; Ultramarine; Purple; Brown; and Achromatic, as seen in Figure 2. Within our pipeline, this new color space was used in the pixel classification section: the RGB input image was converted into CIELAB, and the AB values were translated into polar coordinates (radius (*r*) and angle (θ)). Each pixel was then classified into one of the 12 categories with these

coordinates. The categories were consistent with the unbroken areas obtained using angle bisectors over the color bases. This classification depended solely on the angle for the first 10 categories, the radius for the neutral category, and the angle and radius for the Brown category.

5.2. Fuzzy Color Space Formalization and Color Naming

As described in the previous section, using the angle bisectors on the double AB histogram, we were able to successfully detect and separate different chromogens, accurately segmenting the detected colors. In this Section, to approach the goal of defining an unbiased continuous space for color naming and classification, we applied the principles of fuzzy models applied to multi-labeling systems: in this way, we used trapezoidal heterogeneous membership functions to represent fuzzy colors. Namely, we defined areas of absolute chromogen certainty as the points lying on top of the previously established bases. Moving away from the pure bases, the membership degree to them decreased as we defined increasing gradients of membership for the colors on both sides. The approach was inspired by [8,39–42,48,49].

We defined the set of membership functions using polar coordinates, as shown in Figure 1B, and trapezoidal piecewise polynomials, to maintain continuity, simplifying the problem and matching the geometric angular analysis. For the angular membership, linear piecewise functions were fitted for each color class, so that they would output 1 when the angle of the pixel was in the middle (or peak) of the color class, 0.5 in the boundaries, and 0 when over the peak of the next or previous color class, as given by

$$f(\theta, a, b, c, d, e, g) = \begin{cases} 0 & \text{if } \theta \leq a, \\ \frac{\theta - a}{2(b - a)} & \text{if } a < \theta < b, \\ \frac{\theta - 2b + c}{2(c - b)} & \text{if } b < \theta < c, \\ 1 & \text{if } c < \theta < d, \\ \frac{\theta - 2e + d}{2(d - e)} & \text{if } d < \theta < e, \\ \frac{\theta - g}{2(e - g)} & \text{if } e < \theta < g, \\ 0 & \text{if } \theta > g, \end{cases}$$
(1)

where *a* was the peak for the previous color class, *b* and *e* were the angles corresponding to the boundaries, *c* and *d* were the angles adjacent to the peak of that color class, and *g* corresponded to the peak of the next color class.

This was done using the geometric knowledge learned from the AB reduced space. The radius membership functions were created to differentiate between achromatic (scale of grays), near-achromatic (brown), and pure colors. The function defining the achromatic area was modeled with two linear splines, as given in

$$f_{Achromatic}(r, r_1) = \begin{cases} 1 & \text{if } r \le r_1, \\ 0 & \text{if } r > r_1. \end{cases}$$

$$\tag{2}$$

while three linear splines were used to define the near-achromatic area:

$$f_{Near-Achromatic}(r, r_1, r_2', r_3) = \begin{cases} 0 & \text{if } r \leq r_1, \\ 1 & \text{if } r_1 < r < r_2', \\ \frac{r-r_3}{r_2'-r_3} & \text{if } r_2' < r < r_3, \\ 0 & \text{if } r \geq r_3, \end{cases}$$
(3)

10 of 23

and the chromatic area:

$$f_{Chromatic}(r, r_2', r_3) = \begin{cases} 0 & \text{if } r \le r_2', \\ \frac{r - r_2'}{r_2' - r_3} & \text{if } r_2' < r < r_3, \\ 1 & \text{if } r \ge r_3. \end{cases}$$
(4)

In the above equations, r_1 is the Critical Achromatic Radius, r_2' is the Critical Near-Achromatic Radius, and r_3 is the Modified Critical Chromatic Radius, obtained from the skeletonized histogram.

It can be seen from Figure 2 that, while the splines for the near-achromatic and achromatic areas showed a similar logic to the angular membership functions—with a clear gradient between the two classes—that was not the case for the achromatic area. To account for this, the Critical Achromatic Radius (r_1) was used, as defined in the previous section: all colors with radii lower than r_1 were classified as achromatic. However, the Critical Near-Achromatic Radius (r_2) was used as the point at which the membership function would output 0.5 for near-achromatic and chromatic. Thus, two new points were created: the Modified Critical Near-Achromatic Radius (r_2) and the Critical Chromatic Radius (r_3), both separated 5 points from the original Critical Near-Achromatic Radius (r_2).

With this approach, we obtained an exhaustive analysis of each shade in the input image. We obtained a list of the 10 main shades from each color category. Each shade was then described as a percentual combination of the main color categories required to create it. Thus, a shade with polar coordinates R = 41, Theta = 8° was named "44% Brown, 36% Light Orange, 20% Yellow", corresponding to a yellowish light beige. We believe that this naming technique, based on mathematical principles, provided an excellent solution to the semantic gap, representing an optimal halfway point between human understanding and computer representation of colors.

To summarize, we made use of concepts from different disciplines in the development of ABANICCO: firstly, we centered our model on color theory, which we used to find the location of colors in space, and to represent and name them accordingly; then, we employed geometric analysis to find the best boundaries between color bases; finally, we combined fuzzy theory and multi-label classification, to create a continuous space. The resulting space was closer to human understanding of color, from which non-exclusive labels could be assigned to each detected color, to name them accurately, thus bridging the semantic gap.

6. Results and Discussion

In this section, we describe the experiments performed to validate our approach. This empirical evaluation provides a qualitative and quantitative verification of ABANICCO's effectiveness beyond providing only a simple visual assessment of its results.

While color analysis is a crucial component in various computer vision tasks, the assessment of analytic models is decidedly troublesome. As explained in Section 2.1, the perception of color is influenced by many subjective factors, such as the individual's age, gender, culture, and personal experiences. While most computer vision algorithms are assessed using human-made labels, in the case of color analysis, these human-made annotations can never be objective: they differ, depending on the group creating them. In addition, color measurement devices can be influenced by factors such as lighting conditions, substrate color, and sample size, making it difficult to achieve consistent and reliable results. Moreover, in most applications, color is often considered an attribute of an object rather than a standalone concept: this distinction, along with the challenges posed by human perception, context, and interpretation, results in a scarcity of ground truth data and adequate analysis metrics for color analysis as an individual concept.

To address these difficulties, we designed two types of experiment. The first type of experiment quantitatively evaluated ABANICCO's accuracy in color classification and color naming, using as a benchmark the ISCC–NBS color classification Level 2 (presented

in Section 6.1). The second class of experiment explored the use of ABANICCO in colorbased image segmentation (presented in Section 6.2). We firstly conducted a qualitative evaluation of the performance of ABANICCO's clustering-based color segmentation pipeline against the state-of-the-art fuzzy-based color segmentation algorithm (presented in Sections 6.2.1 and 6.2.2); then, we quantitatively assessed the validity of this pipeline for object segmentation, using the ITTP-Close dataset (presented in Section 6.2.3). Thanks to this empirical evaluation, we provided a numerical comparison of the proposed model against state-of-the-art benchmarks.

6.1. Classification and Naming

To numerically measure ABANICCO's accuracy for the tasks of color classification and naming, we applied our model to the official ISCC–NBS color classification Level 2 [26], established in 1955 by the Inter-Society Color Council and the National Bureau of Standards. This second level of the ISSCC–NBS color system includes a set of 29 colors made up of 10 hues, three neutrals, and 16 compound color names.

We fed our model the image seen in Figure 3A, where the 29 Level 2 ISSCC–NBS prototypes were represented as ordered squares of the same size. ABANICCO classified each of the colors into its 12 categories (Figure 3B), and provided a clear name for each of the individually detected colors. Figure 3 and Table 1 summarize the model's classification and naming performance. Specifically, Figure 3 reflects the results regarding color classification, while Table 1 reports the naming performance of ABANICCO. This first experiment proved that our method could successfully classify the shades into the expected categories, and provide complete, coherent labeling for the ISCC-NBS system names. However, some of the terms in the ISCC–NBS system were incongruous from a color-wheel perspective [28]: for example, "Yellowish pink" and "Reddish Purple", which combined two non-adjoining categories, skipping the middle "Red" and "Pink" sections. Our method solved these nonlinearities by classifying those colors within the skipped middle categories. These two scenarios (marked in bold in Table 1) were the only ones reporting discrepancies between our method and the standardized ISCC-NBS system. Thus, the quantitative evaluation of the segmentation and classification accuracies of ABANICCO told us that, initially, ABANICCO had 93.1% classification accuracy, which increased to 100% when we disregarded the mentioned nonlinearities.



Figure 3. Evaluation of our method with ISCC–NBS system of color categories: (**A**) Level 2 of the ISCC–NBS system, with each color's name superimposed over its corresponding square; (**B**) ABAN-ICCO classification of image (**A**) into the resulting 12 categories.

ISCC–NBS System Name	Algorithm's Classification	Algorithm's Full Description
Red (R)	Red	86.09% Red and 13.91% Pink
Yellowish Pink (Ypk)	Red	58.77% Brown, 40.04% Red, and 1.19% Pink
Reddish Orange (Ro)	Deep Orange	78.67% Deep Orange and 21.33% Red
Orange (O)	Light Orange	62.69% Light Orange and 37.31% Deep Orange
Orange Yellow (OY)	Light Orange	93.41% Light Orange and 6.59% Yellow
Yellow (Y)	Yellow	55.69% Yellow and 44.31% Light Orange
Greenish Yellow (Gy)	Yellow	97.24% Yellow and 2.76% Green
Olive (Ol)	Yellow	97.61% Yellow and 2.39% Light Orange
Olive Green (Olgr)	Green	74.38% Green and 25.62% Yellow
Yellow-green (YG)	Green	64.75% Green and 35.245% Yellow
Green (Ğ)	Green	62.73% Green and 37.27% Teal
Yellowish Green (Yg)	Green	98.27% Green and 1.73% Teal
Bluish Green (Bg)	Teal	86.26% Teal and 13.71% Green
Greenish Blue (Ğb)	Teal	75.88% Teal and 24.12% Blue
Blue (B)	Blue	88.12% Blue and 11.78% Ultramarine
Purplish Blue (Pb)	Ultramarine	100% Ultramarine
Violet (V)	Purple	57.27% Purple and 42.73% Ultramarine
Purple (P)	Purple	73.47% Purple and 26.53% Pink
Pink (P)	Pink	74.31% Pink and 25.69% Red
Reddish Purple (Rp)	Pink	100% Pink
Purplish Red (Pr)	Pink	82.42% Pink and 17.58% Red
Purplish Pink (Ppk)	Pink	85.01% Pink and 14.99% Purple
Brown (Br)	Brown	77.72% Brown, 7.16% Light Orange, and 15.12% Deep Orange
Reddish Brown (Rbr)	Brown	50.10% Brown, 8.28% Deep Orange, and 41.62% Red
Yellowish Brown (Ybr)	Brown	69% Brown and 31% Light Orange
Olive Brown (Olbr)	Brown	64.40% Brown, 26% Light Orange, and 9.60% Yellow
White (Wh)	Neutral	67.50% Yellow and 32.50% Light Orange
Gray (Gy)	Neutral	67.50% Yellow and 32.50% Light Orange
Black (Bk)	Neutral	67.50% Yellow and 32.50% Light Orange

Table 1. Full analysis of each shade of the official ISCC–NBS color classification Level 2.

6.2. Segmentation

Following the validation of ABANICCO's accuracy in classification and naming, we assessed the effectiveness of our model's color segmentation capabilities. We performed two sets of experiments: Sections 6.2.1 and 6.2.2 focused on color segmentation, and Section 6.2.3 focused on image segmentation. As noted earlier, ABANICCO segments color without human intervention, thereby eliminating the possibility of human bias in color identification: however, this meant that a reliable color segmentation ground truth for comparison was unattainable. The available ground truths were created for image segmentation, and took color as an attribute of the objects in the image. These supervised methods for the segmentation of color as an attribute were subject to the influence of perception and preconceived notions of those objects. By contrast, considering color as a complete entity meant that each pixel held significance, and a more exhaustive range of "perceived colors" could be identified.

This distinction came to light in our second experiment (Section 6.2.1), in which we assessed the color segmentation abilities of ABANICCO, using an image of the flag of the United States of America. Flags are often used to test color segmentation algorithms, because they were originally designed using a limited number of easily recognizable and contrasting shades: these characteristics make it relatively straightforward to create a ground truth from images of flags. Per contra, this simplicity introduces a high context bias, as the annotator knows what they are looking for, and what the annotations are expected to look like—in this case, whole sections of one color. However, in natural images, variations in lighting and movement can alter the original hues, saturation, and intensity of the expected colors: for example, a red flag section may exhibit brown regions, while white areas will reflect the vibrant colors of neighboring sections—variations that are rarely reflected in human annotations. In fact, in our fourth experiment (Section 6.2.3), we explored how similar variations can occur with 3D objects of vibrant colors.

A color-analysis-focused method, such as ABANICCO, will detect these discrepancies in individual pixels, as the purpose is not accurate object segmentation; by contrast, methods built for object segmentation, based on annotations of colors as attributes, will not take notice of these variations. The particular task will thus define the adequacy of the method. Nevertheless, in order to provide a quantitative comparison of ABANICCO to the state-of-the-art methods, we incorporated a simple step of class remodeling into the output provided by ABANICCO, resulting in accurate image segmentation.

6.2.1. Comparison with Fuzzy Color Spaces—Discrete Colors

For this second experiment, we compared ABANICCO with the most representative method of the state of the art for color classification. In their work, M. Mengíbar-Rodrigez et al. [42] used Fuzzy Logic and space optimization to obtain a suitable set of prototypes upon which to build a fuzzy color space for each image or set of similar images.

In contrast to ABANICCO, their method required annotations, and did not detect individual colors. Furthermore, in some cases, the user had to decide the number of categories for prototype learning. Their method returned, as output, a map of the membership degree of each pixel to the studied category, which they represented by modifying the rate of transparency of the segmented images. Specifically, they successfully segmented images of country flags into classes corresponding to each flag's original colors. Compared to a human-generated ground truth of 12,000 color samples, they obtained a 0.955 f-score. The ground truth was not created as a complete segmentation of all the areas in the images, but rather as small, 10-pixel samples of consistent colors within the images: this approach prevented the possible variations of the expected colors from being included in the ground truth. Thus, their method was tested against a partial and biased truth.

In this section, we quantitatively compared the segmentation given by ABANICCO directly to their results, as opposed to using their ground truth, as it is not publicly available. The assessment of segmentation performance is most often determined through the use of quantitative metrics, such as object overlap and distance analysis: however, we argue that such approaches would fail to fully capture the nuances inherent in the membership maps obtained in [41]. For lack of an adequate, problem-aware, quantitative metric that could compare such different outputs [50], we preferred to assess the differences in performance quantitatively in Figure 4.

We compared the performance of both methods, using the image of the United States of America flag seen in Figure 4A. The image selected presented clear boundaries between colors. Moreover, the colors were arranged in clusters far away from each other in the color wheel, which can be appreciated in the polar description of the image in Figure 4G: this image is a prime example of how lighting and movement can affect the colors of an object, creating browns on the shaded areas of the red stripes, and new tones of blue as a result of dark-blue areas reflecting over the white stripes and stars. Nevertheless, the phenomenon of color constancy, human bias, and pre-existing notions about the flag will lead most observers to detect only three colors. These variations of the expected colors are marked in Figure 4, rows E and D.

As ABANICCO is designed for color segmentation and classification, its initial segmentation may not align with the ground truth, as it prioritizes accurate color recognition over object segmentation. In this case, it was clear that the segmentation errors were actually a result of ABANICCO's focus on legitimate color classification: the areas in which either blue or red were reflected onto the white stripes were classified as blue and red, their true colors. The actual color of these areas can be seen marked with an orange and green star, respectively, in the full colors summary of Figure 4E. Therefore, the segmentation using ABANICCO was accurate in terms of color but not in terms of object edges or expected human segmentation. To achieve the expected segmentation, we used the information provided by the shade classification (Figure 4E) and polar description (Figure 4G) to easily modify the class boundaries and classes, so as to best match the application (Figure 4F): this simple step of boundary remodeling allowed us to obtain the expected (biased) segmentation for this application, by controlled, informed decisions based on color theory. When we compare our modified segmentation with the one obtained using [41], we can see that the baseline method had problems classifying the different shades of blue, and also the red color, when it appeared darker due to lighting and movement (Figure 4H). This was most patent in the upper right corner of the image. The use of ABANICCO with an added class remodeling step outperformed the benchmark method in segmenting the expected red, blue, and white areas of the flag.



Figure 4. Comparison of segmentation by our method (ABANICCO) versus fuzzy method in [41]: (A) original natural image with a flag waving; (B) fuzzy color space built from extracted prototypes, using the fuzzy method; (C) row of results (red, blue, and white mapping) of the segmentation, using supervised prototypes with the proposal in [41]; this technique used membership degrees to represent the segmentation; thus, lower saturation corresponded to lower membership certainty; (D) row of results (red, blue, and white category) of the segmentation using ABANICCO; the green and orange squares show areas where our method failed to segment the stripes accurately; (E) colors within the red, blue, and light categories found by ABANICCO; the stars mark the color of the areas of failed segmentation in (D); the green (orange) star marks the color of the area within the green (orange) rectangle; (F) row of segmentation results using ABANICCO after adjusting the boundaries; (G) polar description of the image with the original boundaries in gray and the revised boundaries shown with dotted black lines; (H) overlap of the masks obtained in row F and those obtained by [42]; the different shades of pink indicate where the method in [42] output an uncertain membership to that class—the stronger the pink, the lower the membership. In these images, we can see how their method fails in areas where the differences in illumination changed the expected colors: dark red (brown) and dark white (gray).

Thus, our method can easily go from pure color classification to contextual image segmentation, by allowing the user to make informed decisions concerning class boundaries.

In Section 6.2.3, we further explored ABANICCO's capabilities for the problem of object segmentation, performing in-depth quantitative analysis for this specific problem.

6.2.2. Comparison with Fuzzy Color Spaces—Continuous Colors

The image used in the previous experiment is easy to segment for clustering-based methods such as [41], due to the limited number of colors present, and their distinctive distribution within the color wheel; moreover, obtaining a ground truth of the expected categories within it is relatively easy: to address this, we repeated the evaluation performed in Section 6.2.1 on an image with no clear edges: the whole spectrum of hues of the color wheel, and different shades and tones of said hues. The image in Figure 5A is an excellent example of how the absence of precise edges and context impedes the creation of accurate ground truths for the analysis of true color segmentation. A human annotator would have to decide the number of classes of color present in the image, the exact boundaries between these classes, and whether the differences in value and saturation would affect these limits, thus making the ground truth dependent on the individual. As explained in the previous section, due to the lack of a precise measurement methodology for color segmentation, especially in images with no clear edges, the comparison between the two models in this experiment was limited to a visual assessment.

We applied the method presented in [41] and ABANICCO to the image shown in Figure 5A: the comparison between both models' results is shown in Figure 5B,C. In this scenario, which presented a more challenging image, our method successfully identified and classified the existing colors into the 12 categories created according to color theory. By contrast, the method proposed in [41], without previous knowledge of the image, failed to create coherent categories, mixing blues with browns, and oranges with greens.



Figure 5. Comparison of segmentation with our method to fuzzy method in [41], using a simplified version of the image shown in Figure 1: (**A**) the middle rectangle shows the unraveled color wheel; the rectangles above (below) add an increasing quantity of black (gray) to represent shades (tones); (**B**) results of classification using fuzzy color spaces with three arbitrary classes; the lack of labels, and the gradual transitions between colors, resulted in membership maps with small areas of absolute membership, and unclear class separations; (**C**) results of color classification using ABANICCO; our method automatically divided the image into 12 classes with clear boundaries, corresponding to the main hues present plus the near-achromatic brown shades and tones and the achromatic white and black (empty for this particular image).

6.2.3. Object Segmentation—ITTP-Close Dataset

In this final experiment, we present a quantitative analysis of ABANICCO'S performance when applied to the problem of object segmentation. To this end, the technique of class remodeling described in Section 6.2.1 was tested on natural images of the ITTP-Close dataset introduced in [51]: this dataset consisted of five images (450×500 pixels) of scenes with a close light source. All the images contained volumetric objects, some of which included highlights. The annotations for the ground truth were done by splitting the images into small subregions with guaranteed color constancy, and manually merging them when the colors were indistinguishable by the human eye.

Figure 6 shows the images in the dataset, and their corresponding ground truth; moreover, it presents an excellent example of why image segmentation based on color as an attribute is not the same as color segmentation. In Figure 6C, we see that in a singular area of the image, as defined by the ground truth, there was a wide range of distinct colors that even differed in hue: for example, the book appeared as mostly uniform, unsaturated greenish blue; however, it also showed pinkish browns from the reflection of the orange piece, and yellowish greens from the reflection of the yellow cylinder. These variations were not considered in the ground truth, as they would have interfered with object segmentation, which shows why using color for image segmentation is not the same as color segmentation.

In Figure 7, we see the polar description of this same image, as provided by ABAN-ICCO. It is interesting to observe that in our polar AB color space, the different colors of an area appear naturally grouped in defined clusters; therefore, by simply moving the class boundaries, we obtain a perfect segmentation of the objects present in the image.

In this experiment, we were testing segmentation of objects with clear boundaries, which were easier to annotate than the previous cases explored in this work. Thus, we could easily test this segmentation pipeline against the method proposed by the authors of the dataset in [51]. In their work, they proposed a region adjacency graph framework with color space projective transform as a pre-processing technique. This method is not automatic, and requires initialization. In their work, the authors used the dataset median Intersection over Union score (dataset-mIoU) for the evaluation of their segmentation, as defined by

dataset-mIoU =
$$\sum_{k=0}^{K} \min(\text{IoU}(S_k^*, \tilde{S}_k), 0.5),$$
 (5)

where *K* was a number of ground truth segments. The quality range was also in [0, 1].

The authors of [51] reported a dataset-mIoU of 0.85 for their method. By contrast, our segmentation pipeline achieved a dataset-mIoU score of 0.96, representing a 13% improvement.



Figure 6. The ITTP-Close experimental dataset: (**A**) RGB images present in the dataset; (**B**) the corresponding groundtruths; (**C**) a closer look at the different colors in one area, as defined by the ground truth.



Figure 7. Segmentation of sample image BONCH-01 of the ITTP-Close experimental dataset, using class remodeling of ABANICCO's detected colors as clustering-based image segmentation.

6.3. Discussion

In Section 6, we conducted an empirical evaluation to assess the color naming, classification, and segmentation performance of our proposed method, ABANICCO, compared to different state-of-the-art methodologies. The reported results show that ABANICCO outperforms the benchmark methods in accuracy and simplicity.

Specifically, the performed experiments can be divided into two different categories: the first category dealt with color classification and naming, while the second dealt with color segmentation. Firstly, we proved that ABANICCO was able to correctly classify the colors on the official ISCC–NBS color classification Level 2. Moreover, for each color, ABANICCO provided a coherent name that was understandable for humans and machines alike, thus solving the inconsistency between human semantics and machine-extracted features—that is, the semantic gap. Secondly, regarding color segmentation, three additional experiments compared ABANICCO's capabilities with state-of-the-art methods: these experiments showed that ABANICCO is the only method that prioritizes color analysis over area or object segmentation. Moreover, the proposed AB polar description of the colors allows the user to create clusters that solve image segmentation tasks easily. With this simple step of class remodeling, our segmentation pipeline achieved improved segmentation performance compared to state-of-the-art methods for object and area segmentation.

Thus, ABANICCO successfully dealt with the common pitfalls of other color analysis approaches. By basing our model on objective color theory, we avoided the danger of subjectivity, inability to handle variations in lighting and noise, and the lack of standardization of other models. Moreover, we ensured the maximum range of detectable colors, by building ABANICCO from CIELAB color space. Finally, ABANICCO is the least complex model compared to the state of the art. This, together with not needing any annotations, makes ABANICCO practical for a wide range of applications, and accessible to researchers, regardless of their field.

The experiments conducted suggest that ABANICCO could be useful in scenarios that involve changing objects, an unknown number of classes, limited data, unclear boundaries, or when the user lacks prior knowledge of the image or color theory. Moreover, its robustness and unsupervised nature make ABANICCO particularly useful in cases where deep learning techniques are unavailable, inefficient, or when labeled data is scarce or human-biased. The non-supervised character of the proposed method also eliminates the risk of human perception and context-driven errors in color analysis. Furthermore, ABANICCO's potential extends beyond standalone use, and could serve as an initial step in more complex deep learning tasks, such as reducing training time, and improving results in a wide range of tasks, like the introduction of semantic information as organized color data [52,53], sampling and registration [54], compression [55], image enhancement and dehazing [56,57], spectral unmixing [58], and soft-labeling [59].

In addition to ABANICCO's proven success in automatic color classification, naming, and segmentation, its efficiency is highlighted by its computational simplicity, allowing it to run quickly even on low-end computers. Our experiments show that ABANICCO was able to analyze the 380×570 pixels image in Figure 4 in less than a second on a mid-end computer (an HP laptop with i5, 2.4 Hz, and 16Gb RAM).

While the proposed method has many strengths, it does have a drawback that we aim to address in future work: the pipeline for image segmentation currently requires manual intervention by the user, specifically in the class remodeling step. Our future work will explore integrating various automatic clustering methods, to automate the process entirely. With this improvement, ABANICCO will become a fully automatic, non-supervised, unbiased option for image segmentation, including video analysis, without relying on human intervention.

7. Conclusions

In this paper, we have proposed a novel method for color classification, naming, and segmentation, called ABANICCO. Conventional color analysis techniques can be biased, due to human perception and context. By contrast, our method is solely based on color theory, space geometry, fuzzy logic concepts, and multi-label systems.

ABANICCO automatically detects the colors present in an image, classifies them according to color theory, and shows how they are organized, using an innovative polar AB color space. Each detected color is then assigned a multi-label name according to irregular membership functions, considering the differences in color perception, successfully bridging the semantic gap. Finally, the classification output of ABANICCO can be used directly for either pure color segmentation or, with the simple step of boundary remodeling, for controlled, informed image segmentation based on color theory.

We evaluated ABANICCO in two phases. Firstly, we assessed its accuracy in color classification and naming, using Level 2 of the ISSC–NBS color classification system, and achieving perfect accuracy. Next, we extended ABANICCO's capabilities to the problem of image segmentation through a simple pipeline, comparing its performance to the state-of-the-art fuzzy segmentation and color-based object segmentation methods, proving that it outperforms other methods, while being much more straightforward and efficient.

ABANICCO represents a substantial advancement in the field of color analysis, and has the potential to be applied in a broad range of real-world applications.

Author Contributions: Conceptualization, methodology, formal analysis, investigation, validation, software, writing—original draft, L.N.-S.; methodology, formal analysis, writing—review and editing, A.L.; conceptualization, methodology, validation, supervision, funding acquisition, writing review and editing, J.P.; conceptualization, methodology, validation, supervision, funding acquisition, writing—review and editing, A.M.-B. All authors have read and agreed to the published version of the manuscript.

Funding: This research was funded by the Ministerio de Ciencia, Innovacción y Universidades, Agencia Estatal de Investigación, under grant PID2019-109820RB, MCIN/AEI/10.13039/501100011033 co-financed by the European Regional Development Fund (ERDF), "A way of making Europe" to A.M.-B. and L.N.-S.

Institutional Review Board Statement: Not applicable.

Informed Consent Statement: Not applicable.

Data Availability Statement: All the data, code, resources, and additional materials used in our research are available in https://github.com/lauranicolass/ABANICCO (accessed on 19 March 2023).

Acknowledgments: We want to thank M. Mengíbar-Rodríguez for her help in the results section, and F. Moreno-Pino for his help with the manuscript revision.

Conflicts of Interest: The authors declare no conflict of interest.

Appendix A

Finding Class Boundaries

To find the approximate areas where colors were located, we used a color wheel complete with hue modifications, and studied its appearance in the reduced AB color space of CIELAB. The double histogram was simplified, and subsequently skeletonized, to find the color bases. In Figure A1, we can see the results of this method, obtaining the color bases by joining the center with the endpoints and with the achromatic and near-achromatic areas, represented as circles with center (0, 0) and radius that of the first and second sets of the branch points.



Figure A1. Complete version of the experiment shown in Figure 2A. Top row: from the color wheel, the AB double histogram was created; then, a simplified AB double histogram was computed and skeletonized, to find the color bases; the center of space and the center of the skeleton were marked with a green dot; the branches and the endpoints were marked with a blue and red point, respectively; the dark and the light orange circles indicate the positions of the achromatic (Scale of Grays) and near-achromatic (Brown) areas, respectively. Bottom row: the boundaries between the located color bases were chosen as each angle bisector; the final plot shows the proposed discrete polar color space.

The easiest way to obtain the boundaries between the located color bases was to find each angle bisector. To avoid overlapping with the generated regions, those chromogen coordinate points with angles smaller than 5° were merged, assuming they would belong to the same region: this is shown in step 3 of Figure 2. We also tested the primary state-of-the-art classification methods against the bisector option: to this end, we used MATLAB's classification learner. We created classes of 100 points laying over each of the proposed chromogen bases from the center to the limits of the reduced AB color space, with the bases extended to occupy the totality of the AB space: this was done to ensure that the classification methods would learn the regions entirely, instead of leaving the edges undefined. With these points, we trained the following state-of-the-art classifiers, using five cross-validation folds: Decision Trees (Fine, Medium, and Coarse); Discriminant Analysis; Naïve Bayes; Support Vector Machines; Nearest Neighbor (Fine, Medium, Coarse, Cosine, Cubic, and Weighted); Kernel Approximations; Ensemble (Boosted Trees, Bagged Trees, Subspace Discriminant, Subspace KNN, and RUSBoosted Trees); and Neural Network (Narrow, Medium, Wide,

Bilayered, and Trilayered). For details on the specifics of how each model was trained, refer to the Mathworks page on classification learner: https://www.mathworks.com/help/stats/ classificationlearner-app.html (accessed on 10 January 2023).

Of all the classifiers, the best results were the Fine and Wide KNN, SVM, and Wide Neural Network classifiers, all of which provided a validation accuracy larger than 0.98, and classified almost all the given points correctly. For comparison, we focused on the best of these: the Weighted KNN (wKNN) and the Wide Neural Network (WNN). The wKNN had a validation accuracy of 99.2% in a training time of 2.3318 s, 10 neighbors, and Euclidean distance metric. The WNN obtained a validation accuracy of 99.6% after training for 1707.9 s with one fully connected layer and ReLu activation. Due to the simplicity of the needed geometry, it was expected that the best models would be the simplest ones, since all we needed to find were radial lines between color bases; however, the resulting regions did not match the expected ones and failed to provide acceptable pixel classification when applied to actual images: specifically, they mixed different colors in the same regions (as seen in comparative Figures A2 and A3). These approaches had good prediction accuracy, but were unsuccessful when creating the expected color regions, especially in the warmer areas of red-orange-pink, and in telling blues and purples apart. This comparative analysis proved that the best regions were those obtained using the geometrical approach of the angle bisectors.



Figure A2. Segmentations obtained after using machine learning methods and the bisector approach to find the boundaries between the color bases. This evaluation was carried out in three rectangles depicting the concept of color wheel shadows and tones. Rectangle 1 shows the full-color wheel. Rectangle 3 above was obtained by adding 75% of black to represent shades. Rectangle 2 below was obtained by adding 75% of gray to represent tones. Above each rectangle are lines measuring the length of the classes obtained by each method. The solid gray lines represent the classification carried out by wKNN. The solid black lines represent the classification by WNN. The color lines represent the classification by our method, with just the bisectors and, above, the simplified final ABANICCO classes. The black rectangle marks the area used for the evaluation in Figure A3.



Figure A3. Closer look at the comparison of color classification by machine learning methods and our method using the area marked in Figure A2. This evaluation was carried out in 7 rectangles depicting the concept of shadows and tones of a portion of the color wheel. Rectangle 1 shows the portion of the color wheel from bluish-green to red. The rectangles above this were obtained by adding an increasing quantity of black to represent shades (Rectangles 2.a, 3.a, and 4.a). The rectangles below were obtained by adding an increasing quantity of gray to represent tones (Rectangles 2.b, 3.b, and 4.b). The figure displays the results of color classification over the region covering the purplish-blue and pinkish-red colors. The region's exact limits and length depended on the shape of the class created by each method. Above each rectangle, three different lines measured the length of the segmentations by each approach, for this portion of the color wheel: the solid gray line corresponded to the k-nearest neighbors algorithm (KNN); the solid black line to the Wide Neural Network (WNN); the tri-color line to our method. A solid color line represents both Machine Learning methods, because they failed to create appropriate boundaries between blues, purples, and pinks. By contrast, our method is represented using three colors, because it obtained three different classes in all the experiments, regardless of tonality. To both sides of these lines are arrows indicating the classes obtained before and after this section; the actual segmentation performed by the three methods on rectangles 4.a, 1, and 4.b is shown to their right. The separation between classes of our method is marked with white lines for clarity.

References

- Ganesan, P.; Sajiv, G. User oriented color space for satellite image segmentation using fuzzy based techniques. In Proceedings of the 2017 International Conference on Innovations in Information, Embedded and Communication Systems (ICIIECS), Coimbatore, India, 17–18 March 2017; pp. 1–6.
- 2. Dadwal, M.; Banga, V.K. Estimate ripeness level of fruits using RGB color space and fuzzy logic technique. *Int. J. Eng. Adv. Technol.* **2012**, *2*, 225–229.
- Huang, Z.; Liu, Q.; Pointer, M.; Luo, M.; Wu, B.; Liu, A. White lighting and colour preference, Part 1: Correlation analysis and metrics validation. *Light. Res. Technol.* 2020, 52, 5–22. [CrossRef]
- 4. Deng, X.; Liu, Y.; Tian, B.; Zhang, W.; Yu, F.; Liu, Q. Experimental setting and protocol impact human colour preference assessment under multiple white light sources. *Front. Neurosci.* **2022**, *16*, 1029764. [CrossRef]
- 5. Foster, D.H. Color constancy. Vis. Res. 2011, 51, 674–700. [CrossRef]

- 6. Mylonas, D.; Caparos, S.; Davidoff, J. Augmenting a colour lexicon. Humanit. Soc. Sci. Commun. 2022, 9, 29. [CrossRef]
- 7. Rosch, E. Cognitive representations of semantic categories. J. Exp. Psychol. Gen. 1975, 104, 192. [CrossRef]
- Chamorro-Martínez, J.; Soto-Hidalgo, J.M.; Martínez-Jiménez, P.M.; Sánchez, D. Fuzzy color spaces: A conceptual approach to color vision. *IEEE Trans. Fuzzy Syst.* 2016, 25, 1264–1280. [CrossRef]
- Yazici, V.O.; Weijer, J.v.d.; Ramisa, A. Color naming for multi-color fashion items. In Proceedings of the World Conference on Information Systems and Technologies, Naples, Italy, 27–29 March 2018; Springer: Berlin/Heidelberg, Germany, 2018; pp. 64–73.
- 10. Tian, K.; Li, J.; Zeng, J.; Evans, A.; Zhang, L. Segmentation of tomato leaf images based on adaptive clustering number of K-means algorithm. *Comput. Electron. Agric.* **2019**, *165*, 104962. [CrossRef]
- Saifullah, S. Segmentation for embryonated Egg Images Detection using the K-Means Algorithm in Image Processing. In Proceedings of the 2020 Fifth International Conference on Informatics and Computing (ICIC), Gorontalo, Indonesia, 3–4 November 2020; pp. 1–7. [CrossRef]
- 12. Helmenstine, A. The Visible Spectrum: Wavelengths and Colors. 2020. Available online: https://www.thoughtco.com/ understand-the-visible-spectrum-608329 (accessed on 18 December 2022).
- 13. Lee, B.B. The evolution of concepts of color vision. Neurociencias 2008, 4, 209.
- 14. Jacobs, G.H. Primate photopigments and primate color vision. Proc. Natl. Acad. Sci. USA 1996, 93, 577-581. [CrossRef]
- 15. Grandy, D. Goethe on color and light. J. Interdiscip. Stud. 2005, 17, 26–44. [CrossRef]
- 16. Westfall, R.S. The development of Newton's theory of color. *Isis* **1962**, *53*, 339–358.
- 17. Mylonas, D.; Griffin, L.D. Coherence of achromatic, primary and basic classes of colour categories. *Vis. Res.* **2020**, *175*, 14–22. [CrossRef] [PubMed]
- 18. Mulholland, R.; Howell, D.; Beeby, A.; Nicholson, C.; Domoney, K. Identifying eighteenth century pigments at the Bodleian library using in situ Raman spectroscopy, XRF and hyperspectral imaging. *Herit. Sci.* **2017**, *5*, 43. [CrossRef]
- 19. Mabberley, D.J. Painting by Numbers: The Life and Art of Ferdinand Bauer; NewSouth Publishing: Montgomery, AL, USA, 2017.
- 20. Parkhurst, C.; Feller, R.L. Who invented the color wheel? Color Res. Appl. 1982, 7, 217–230. [CrossRef]
- 21. MacEvoy, B. Modern Color Models. Watercolors. 2005. Available online: http://www.handprint.com/HP/WCL/color7.html (accessed on 18 December 2022).
- 22. Witzel, C. Misconceptions about colour categories. Rev. Philos. Psychol. 2019, 10, 499–540. [CrossRef]
- 23. McLachlan, F. Visual heuristics for colour design. In *Visual Research Methods in Architecture*; Intellect Books: Bristol, UK, 2020; pp. 335–352.
- 24. Werner, A.G.; Syme, P. Werner's Nomenclature of Colours; W. Blackwood: Edinburgh, UK, 1821.
- 25. Hamly, D. Robert Ridgway's color standards. Science 1949, 109, 605–608. [CrossRef] [PubMed]
- 26. Inter-Society Color Council. ISCC. Available online: https://www.iscc.org/ (accessed on 18 September 2022).
- 27. Berlin, B.; Kay, P. Basic Color Terms: Their Universality and Evolution; University of California Press: Berkeley, CA, USA, 1991.
- 28. Moss, A.E. Basic colour terms: Problems and hypotheses. Lingua 1989, 78, 313–320. [CrossRef]
- 29. Munsell, A.H. A Color Notation: A Measured Color System Based on the Three Qualities, Hue, Values and Chroma with Illustrative Models, Charts and a Course of Study Arranged for Teachers; Press of Geo. H. Ellis Co.: Boston, MA, USA, 1907.
- 30. Munsell, A.H. A pigment color system and notation. Am. J. Psychol. 1912, 23, 236–244. [CrossRef]
- 31. Mylonas, D.; MacDonald, L. Augmenting basic colour terms in English. Color Res. Appl. 2016, 41, 32–42. [CrossRef]
- Pegalajar, M.; Ruíz, L.G.B.; Sánchez-Marañón, M.; Mansilla, L. A Munsell colour-based approach for soil classification using Fuzzy Logic and Artificial Neural Networks. *Fuzzy Sets Syst.* 2020, 401, 38–54. [CrossRef]
- Gershikov, E.; Porat, M. Correlation VS. decorrelation of color components in image compression—Which is preferred? In Proceedings of the 2007 15th European Signal Processing Conference, Poznan, Poland, 3–7 September 2007; pp. 985–989.
- Jin, Y.; Liu, Z.; Wang, P.F.; Liu, S.G.; Zhai, H.J. Study on Correlation of Color Components Image in Different Color Spaces. In Proceedings of the Applied Mechanics and Materials; Trans Tech Publications: Stafa-Zurich, Switzerland, 2013; Volume 262, pp. 86–91.
- 35. Colantoni, P.; Thomas, J.B.; Trémeau, A. Sampling CIELAB color space with perceptual metrics. *Int. J. Imaging Robot.* **2016**, *16*, 1–22.
- 36. Abshire, C.; Allebach, J.P.; Gusev, D.A. Psychophysical study of color verbalization using fuzzy logic. *Electron. Imaging* **2016**, 2016, art00033. [CrossRef]
- 37. Shamir, L. Human Perception-based Color Segmentation Using Fuzzy Logic. IPCV 2006, 2, 96–502.
- Leon-Garza, H.; Hagras, H.; Peña-Rios, A.; Conway, A.; Owusu, G. A big bang-big crunch type-2 fuzzy logic system for explainable semantic segmentation of trees in satellite images using hsv color space. In Proceedings of the 2020 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), Glasgow, UK, 19–24 July 2020; pp. 1–7.
- Chamorro-Martínez, J.; Keller, J.M. Granular Modeling of Fuzzy Color Categories. *IEEE Trans. Fuzzy Syst.* 2019, 28, 1897–1909. [CrossRef]
- Chamorro-Martínez, J.; Mengíbar-Rodríguez, M.; Martínez-Jiménez, P.M. A Path-based Fuzzy Approach to Color Image Segmentation. In Proceedings of the 2021 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), Luxembourg, 11–14 July 2021; pp. 1–6.
- 41. Mengíbar-Rodríguez, M.; Chamorro-Martínez, J. Learning prototypes for building fuzzy color spaces. In Proceedings of the 2022 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), Padua, Italy, 18–23 July 2022; pp. 1–6.

- 42. Mengíbar-Rodríguez, M.; Chamorro-Martínez, J. An image-based approach for building fuzzy color spaces. *Inf. Sci.* 2022, 616, 577–592. [CrossRef]
- 43. Sturges, J.; Whitfield, T.A. Locating basic colours in the Munsell space. Color Res. Appl. 1995, 20, 364–376. [CrossRef]
- 44. Basar, S.; Ali, M.; Ochoa-Ruiz, G.; Zareei, M.; Waheed, A.; Adnan, A. Unsupervised color image segmentation: A case of RGB histogram based K-means clustering initialization. *PLoS ONE* **2020**, *15*, e0240015. [CrossRef]
- 45. Xing, Z. An improved emperor penguin optimization based multilevel thresholding for color image segmentation. *Knowl. Based Syst.* **2020**, *194*, 105570. [CrossRef]
- 46. Bao, X.; Jia, H.; Lang, C. A novel hybrid harris hawks optimization for color image multilevel thresholding segmentation. *IEEE Access* **2019**, *7*, 76529–76546. [CrossRef]
- 47. Liang, H.; Jia, H.; Xing, Z.; Ma, J.; Peng, X. Modified grasshopper algorithm-based multilevel thresholding for color image segmentation. *IEEE Access* **2019**, *7*, 11258–11295. [CrossRef]
- Chamorro-Martínez, J.; Marín, N.; Mengíbar-Rodríguez, M.; Rivas-Gervilla, G.; Sánchez, D. Referring expression generation from images via deep learning object extraction and fuzzy graphs. In Proceedings of the 2021 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), Luxembourg, 11–14 July 2021; pp. 1–6.
- Forcén, J.I.; Pagola, M.; Bustince, H.; Soto-Hidalgo, J.M.; Chamorro-Martínez, J. Adding fuzzy color information for image classification. In Proceedings of the 2017 IEEE International Conference on Fuzzy Systems (FUZZ-IEEE), Naples, Italy, 9–12 July 2017; pp. 1–6.
- 50. Maier-Hein, L.; Menze, B. Metrics reloaded: Pitfalls and recommendations for image analysis validation. *arXiv* 2022, arXiv:2206.01653.
- Smagina, A.; Bozhkova, V.; Gladilin, S.; Nikolaev, D. Linear colour segmentation revisited. In Proceedings of the Eleventh International Conference on Machine Vision (ICMV 2018), Munich, Germany, 1–3 November 2018; International Society for Optics and Photonics: Bellingham, WA, USA, 2019; Volume 11041, p. 110410F.
- 52. Bhowmick, A.; Saharia, S.; Hazarika, S.M. Non-parametric scene parsing: Label transfer methods and datasets. *Comput. Vis. Image Underst.* **2022**, *219*, 103418. [CrossRef]
- 53. Wheeler, B.J.; Karimi, H.A. A semantically driven self-supervised algorithm for detecting anomalies in image sets. *Comput. Vis. Image Underst.* **2021**, 213, 103279. [CrossRef]
- Saval-Calvo, M.; Azorin-Lopez, J.; Fuster-Guillo, A.; Villena-Martinez, V.; Fisher, R.B. 3D non-rigid registration using color: Color coherent point drift. *Comput. Vis. Image Underst.* 2018, 169, 119–135. [CrossRef]
- 55. Baig, M.H.; Torresani, L. Multiple hypothesis colorization and its application to image compression. *Comput. Vis. Image Underst.* **2017**, *164*, 111–123. [CrossRef]
- Jiang, Y.; Sun, C.; Zhao, Y.; Yang, L. Image dehazing using adaptive bi-channel priors on superpixels. *Comput. Vis. Image Underst.* 2017, 165, 17–32. [CrossRef]
- 57. El Khoury, J.; Le Moan, S.; Thomas, J.B.; Mansouri, A. Color and sharpness assessment of single image dehazing. *Multimed. Tools Appl.* **2018**, *77*, 15409–15430. [CrossRef]
- Grillini, F.; Thomas, J.B.; George, S. Comparison of Imaging Models for Spectral Unmixing in Oil Painting. Sensors 2021, 21, 2471. [CrossRef]
- 59. Sucholutsky, I.; Schonlau, M. 'Less than one'-shot learning: Learning N classes from M<N samples. arXiv 2020, arXiv:2009.08449.

Disclaimer/Publisher's Note: The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.